

# 

**Citation:** Abbate JL, Ezenwa VO, Guégan J-F, Choisy M, Nacher M, Roche B (2018) Disentangling complex parasite interactions: Protection against cerebral malaria by one helminth species is jeopardized by co-infection with another. PLoS Negl Trop Dis 12(5): e0006483. https://doi. org/10.1371/journal.pntd.0006483

Editor: Matthew C. Freeman, Emory University, UNITED STATES

Received: December 29, 2017

Accepted: April 30, 2018

Published: May 10, 2018

**Copyright:** © 2018 Abbate et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** The datasets supporting this article have been uploaded as part of the supplementary material.

**Funding:** This study is a contribution of the ANR JC project 'STORY', granted to BR and supporting JLA. MN, JFG and BR are supported by 'an Investissement d'Avenir' grant from the Agence Nationale de la Recherche (CEBA ANR-10-LABX-2501). VOE received support from a US Fulbright Scholar Award. The funders had no role in study

RESEARCH ARTICLE

Disentangling complex parasite interactions: Protection against cerebral malaria by one helminth species is jeopardized by coinfection with another

Jessica L. Abbate<sup>1,2</sup>\*, Vanessa O. Ezenwa<sup>3</sup>, Jean-François Guégan<sup>2</sup>, Marc Choisy<sup>2,4</sup>, Mathieu Nacher<sup>5,6</sup>, Benjamin Roche<sup>1,2,7</sup>

 UMMISCO, IRD / Sorbonne Université, Bondy, France, 2 MIVEGEC, IRD, CNRS, Université Montpellier, Montpellier, France, 3 Odum School of Ecology and Department of Infectious Diseases, College of Veterinary Medicine, University of Georgia, Athens, GA, United States of America, 4 Oxford University Clinical Research Unit, Hanoi, Vietnam, 5 CIC INSERM 1424, Centre Hospitalier de Cayenne, Cayenne, French Guiana, 6 EA3593, Ecosystèmes Amazoniens et Pathologie Tropicale, Université de Guyane, Cayenne, French Guiana, 7 Departamento de Etología, Fauna Silvestre y Animales de Laboratorio, Facultad de Medicina Veterinaria y Zootecnia, Universidad Nacional Autónoma de México (UNAM), Ciudad de México, México

\* jessie.abbate@gmail.com

## Abstract

Multi-species interactions can often have non-intuitive consequences. However, the study of parasite interactions has rarely gone beyond the effects of pairwise combinations of species, and the outcomes of multi-parasite interactions are poorly understood. We investigated the effects of co-infection by four gastrointestinal helminth species on the development of cerebral malaria among *Plasmodium falciparum*-infected patients. We characterized associations among the helminth parasite infra-community, and then tested for independent (direct) and co-infection dependent (indirect) effects of helminths on cerebral malaria risk. We found that infection by *Ascaris lumbricoides* and *Trichuris trichiura* were both associated with direct reductions in cerebral malaria risk. However, the benefit of *T. trichiura* infection was halved in the presence of hookworm, revealing a strong indirect effect. Our study suggests that the outcome of interactions between two parasite species can be significantly modified by a third, emphasizing the critical role that parasite community interactions play in shaping infection outcomes.

## Author summary

Over the last decade, an increasing number of studies have shown that co-infected individuals can show strikingly different disease symptoms than singly infected individuals. Among these studies, the potential protective effect of some helminth species on malaria severity has attracted considerable attention because of the possible public health impact. However, despite the fact that helminth species frequently co-occur together within an individual, no studies have investigated the role of within-host helminth community interactions on malaria severity. Taking the advantage of a cohort study in Thailand, we design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

found that infection by *Ascaris lumbricoides* and *Trichuris trichiura* were both associated with direct reductions in cerebral malaria risk. However, the benefit of *T. trichiura* infection was halved in the presence of hookworm, revealing that interactions among helminth species can jeopardize the beneficial role of individual helminths. Our study shows that the outcome of interactions between two parasite species can be significantly modified by a third, emphasizing the critical role that parasite community interactions play in shaping infection outcomes and their impact on public health. These insights are especially relevant for neglected tropical diseases that occur in areas of high pathogen diversity.

### Introduction

Most hosts are co-infected by a range of pathogenic organisms (e.g., [1]), and interactions between co-infecting parasites can be important determinants of susceptibility to infection (e.g., [2]) as well as the severity of disease symptoms (e.g., [3]). Co-occurring parasites interact within a host via a number of mechanisms, including through active down-regulation of immune responses by some parasites, indirect competition through stimulation of host defenses, and direct competition for resources [4]. Importantly, the resulting impacts of within-host parasite interactions on host health can be beneficial (e.g., cross-protection between influenza strains, [5]) or detrimental (e.g., stimulation of the immune system's response against many helminth species reduces the capacity of vertebrate hosts to respond to some intracellular pathogens [6]).

The majority of ecological studies on co-infection have focused on understanding the ecological, epidemiological, and evolutionary consequences of pairwise parasite interactions (e.g., [7-12]). However, since the mechanisms by which any two parasites can interact are not mutually exclusive, the costs of co-infection to host health may frequently be offset by simultaneous benefits or influenced by other co-infecting parasites [13]. In community ecology, a rich body of literature on direct and indirect effects of multi-species interactions indicates that interactions between multiple species in a community can be additive, non-additive, or lead to entirely non-intuitive "ecological surprises" (e.g., [14–18]). Yet, remarkably few studies have looked beyond direct pairwise parasite-parasite interaction effects on disease (but see e.g., [19– 21]).

Interactions between helminths and human malaria (particularly that caused by *Plasmo-dium falciparum*) are among the most well-studied pairwise parasite interactions [22]. Both experimental studies in rodents and cross-sectional studies in humans suggest that immuno-suppressive effects of some helminth species can weaken host defenses against *Plasmodium spp.*, resulting in increased malaria incidence and severity, while co-infection with other helminth species is associated with protective effects against cerebral malaria, the most lethal form of malaria (reviewed in [23,24]). However, in addition to interactions between helminths and *Plasmodium*, different helminth species can also affect one another [25,26]. For example the trematode *Schistosoma mansoni* has been shown to impair normal clearance of *Echinostoma revolutum* (another trematode) in experimentally co-infected lab mice [27]. In humans, there is also correlative evidence that *A. lumbricoides* and hookworm, two soil-transmitted intestinal helminths known to interact with *Plasmodium*, may themselves interact [28]. Thus, it is possible that the effects of any single helminths in the system.

Given the possible web of complex interactions that can occur in the helminth-*Plasmodium* system, we tested the hypothesis that interactions between helminth species can modify the

impact of single-species helminth infections on malaria severity. Taking advantage of a dataset on 283 *Plasmodium*-infected patients from western Thailand, who were screened for helminth infections and monitored for the development of cerebral malaria [29,30], we combined standard logistic regression analysis with an association screening approach (SCN; [31]), to characterize how both direct (independent) and indirect (co-infection dependent) effects of helminth species influence malaria severity. To do this, we first identified and quantified the types of coinfections occurring between four different helminth species in our study population, and then tested for significant interspecific associations. Next, we evaluated the impact of single helminths, and their interactions with one-another, on the development of cerebral malaria. By taking advantage of the well-described helminth-malaria co-infection system to investigate how helminths interact with malaria severity both singly and in combination, we provide new insight into the consequences of parasite interactions.

#### Materials and methods

#### Patient data

Passive surveillance data on malaria patients were collected from the Hospital for Tropical Diseases, Bangkok, Thailand, and are described in detail elsewhere [29,30]. The study was approved by the Ethical Committee of the Faculty of Tropical Medicine, Mahidol University and all data analyzed were anonymized [30]. A total of 283 individuals with hyperparasitemic malaria (high P. falciparum biomass, defined as parasitemia >5% or >200,000/µl, or presence of schizonts in a peripheral bloodsmear) were screened for the development of cerebral malaria (neurological dysfunction typically leading to death or disability) and the presence of four helminth species: Ascaris lumbricoides (roundworm, AL), Trichuris trichuira (whipworm, TT), hookworm (HW, primarily *Necator americanus*, but also including other species such as Ancylostoma duodenale), and Strongyloides stercoralis (threadworm, SS) in fecal samples. Experienced microscopists were responsible for screening for both *Plasmodium* (using thick and thin blood smears stained with 10% Giemsa and read at 1000x magnification) and gastrointestinal nematode species via egg morphology (also using standard light microscopy techniques). Sex and age class (<20 years old, 20-40 years old, and >40 years old) were also recorded. Out of the 283 hyperparasitemic cases, 67 developed cerebral malaria, while the remaining 216 did not. The demographic composition of the study population is provided in Table 1.

#### Overview of the analytical approach

Given that co-infection by multiple helminth species commonly occurs, univariate analysis of each species' impacts on the development of cerebral malaria may be confounded by the

		Cerebral malaria	Hyperparasitemic controls		
Sex					
	Female	14 (20.9%)	62 (28.8%)		
	Male	55 (79.1%)	153 (71.2%)		
Age					
	Median [range]	25 [14–72]	23 [3-74]		
	< 20	15 (22.3%)	73 (33.8%)		
	20-40	44 (65.7%)	123 (56.9%)		
	>40	8 (11.9%)	17 (7.9%)		

**Table 1. Demographic composition of study population.** There were slightly more females in the <20 age group (35%) than in those 20–40 (23%) or over 40 (24%), but this was not statistically significant ( $\chi^2$  = 4.3, df = 2, p = 0.12).

https://doi.org/10.1371/journal.pntd.0006483.t001

patient's co-infection status. Furthermore, facilitative or competitive interactions among helminth species would mean that occurrences of interacting parasites in any given patient are not independent of one-another. To address this issue, we first identified co-infection frequencies and quantified associations between helminth species using association screening (SCN) analysis, a method recently developed to identify complex association patterns in large parasite assemblages [31], combined with a rarefaction analysis. This approach allowed us to understand how single and multiple-species infections were distributed among patients and to infer the relative strength of associations identified as significant. We then tested for direct and indirect effects of each helminth species on the development of cerebral malaria using logistic regression analysis, correcting for co-infection status with analysis of deviance and explicitly including all possible interactions terms in the model. We used a model selection approach to identify the helminth interactions that had significant consequences on the development of cerebral malaria.

## Quantifying associations between helminth species

To identify potential interactions occurring between all possible combinations of the four helminth species in our dataset, we used SCN analysis [31] to quantify the absolute and relative strength of species associations in the dataset. The utility of this method is that in addition to identifying multi-species associations with a high degree of power, it identifies the precise nature of these associations (i.e. whether it is the presence or absence of each species that occurs more often than expected by chance). The absolute strength of an association was determined by the SCN analysis itself, applied as a p-value (described below). As strong associations are likely to have identical p-values, the relative strength of an association was determined by running the SCN analysis on systematically jackknifed subsets of the data (also described below), with the assumption that the detection of stronger associations will be less influenced by sampling biases. Given the prevalence of each parasite species in the study population, SCN analysis generates a simulation-based 95% confidence envelope around the expected frequency of each possible combination of concurrent infection status (a total of  $2^{NP}$  combinations, where NP = the number of parasite species) under the null hypothesis of random parasite associations. Observed coinfection combinations falling above or below this envelope are considered to occur more or less frequently, respectively, than in 95% of the random simulations. Significance of the association is given as a *p*-value, calculated as the number of instances in which the simulated co-infection frequency differed (above or below the upper or lower threshold, respectively) from the observed frequency divided by the total number of simulations, and is thus dictated by the variance in the simulated frequencies. Power tests on this method revealed that there should be at least as many host individuals in the dataset as the number of possible parasite species combinations  $(2^{NP})$ , and that stronger associations had lower *p*-values [31].

We tested the robustness of each association to sampling biases by rarefaction analysis, where the SCN analysis was performed on progressive sub-sets of the data: 10 partitions sampling from 10% to 100% of the data (see Supplementary Material S1 Appendix). For each partition, the SCN analysis was performed on 20 independent sub-samples; and associations were deemed 'significant' if the *p*-value was below 0.05 in at least 95% of the runs. A robustness score (RR) was given to each species combination, ranging from 0 to 10, representing how much data sub-setting an association could tolerate before it was no longer detectable. A score of RR = 1 indicates that 100% of the data are needed, RR = 10 indicates that the association was detected in all 10 partitions. Stronger associations will have higher RR scores; thus, RR scores provide a quantitative measure of the relative strength of significant associations which often have similar *p*-values.

## Examining direct and indirect effects of helminths on cerebral malaria

Direct effects of infection by each helminth species on the development of cerebral malaria, and indirect effects of interactions between helminth species, were tested using logistic regression with backwards stepwise removal of higher-order interaction terms. Analyses were implemented using the 'glm' function from the *stats* package in R [32] with a binomial logit link. The 'step' function was used for backward model selection to reduce the number of interactions in the final model by minimizing the relative Akaike information criterion (AIC) value. On this final model,  $\chi^2$  analysis of deviance was performed to test each helminth species' impact on cerebral malaria by first correcting for patient sex, age class, and co-infection status (using the 'drop1' function).

## Results

## Helminth infection and co-infection frequencies

Out of the 283 hyperparasitemic malaria patients screened, 139 (49.1%) were infected with at least one helminth species, including 20 (29.8%) of 67 that developed cerebral malaria. The most prevalent helminth infection was HW (infecting 30.1% of all patients), followed by TT (24.7%), AL (19.1%), and SS (10.6%). Half (49.6%) of all patients infected with helminths were infected with more than one helminth species, including 4 individuals who were co-infected with all four species (Table 2). The distribution of these co-infections among patients that did or did not develop cerebral malaria is given in S1 Table.

## Associations between helminth species

A number of significant associations between helminths were identified via SCN and rarefaction analysis. The most robust result was that three-way co-infections between AL, TT, and

**Table 2.** Association screening (SCN) analysis results for all detected gastrointestinal helminth species infecting hyperparasitemic malaria patients. The observed (Obs) frequency of each co-infection status is given along with the lower (LL) and upper (UL) limits of the 95% confidence envelope. Robustness scores are given for significant associations (SCN p<0.05 for at least 95% of runs, see main text). SCN *P*-values reported are the maximum *p*-value returned by the SCN analysis when 100% of the data are sampled. Significant or trending associations are highlighted in bold for emphasis.

SCN and Rarefaction Results				All hyperparasitemic malaria patients (N = 283)													
<b>Coinfection Status</b>	18 Pathogen species present			Obs	LL	UL	Direction	Percentage of data sampled					mpled		Robustness Score	SCN p-value < =	
									<b>10%</b> → <b>100%</b>								
4 species	AL	ТТ	HW	SS	4	0	3	Frequent						✓	✓	2	0.0032
3 species	AL	TT	HW		17	0	10	Frequent	✓	✓	~	✓	✓	✓	✓	7	1.0 x 10 <sup>-8</sup>
	AL	TT		SS	1	0	5	Rare								0	0.6410
	AL		HW	SS	2	0	5	Rare								0	0.7676
		TT	HW	SS	3	0	7	Random								0	0.5556
2 species	AL	TT			11	1	18	Frequent								0	0.4520
	AL		HW		9	3	21	Rare								0	0.7276
	AL			SS	1	0	9	Rare								0	0.4120
		TT	HW		14	5	28	Rare								0	0.9068
		TT		SS	1	0	11	Rare								0	0.1692
			HW	SS	6	0	14	Random								0	0.9600
Single infections	AL				9	13	41	Rare			~	~	✓	✓	✓	5	0.0012
		ТТ			19	20	52	Rare				✓	✓	✓	✓	4	0.0036
			HW		30	30	64	Rare					✓	✓	✓	3	0.0132
				SS	12	4	24	Rare								0	0.9960
Not infected					144	84	132	Frequent	✓	~	1	~	~	1	~	7	0.0004

https://doi.org/10.1371/journal.pntd.0006483.t002

HW, occurred more often than expected by random chance (p < 0.00001, RR = 7), indicating a strong positive 3-way association between these species (Table 2). Uninfected individuals were also highly over-represented in the dataset (p = 0.0004, RR = 7), while AL, TT, and HW were all found to occur alone significantly less often than expected by chance (Table 2). The weakest significant result suggested that all four parasites (Al, TT, HW, and SS) were involved in more frequent co-infection than expected by chance (p = 0.0032, RR = 2; Table 2). However, given the relative strength of the 3-way association between AL, TT, and HW and the random occurrence of solo-SS infections (p = 0.996, RR = 0), it is possible this result simply indicates that the mechanism responsible for the 3-way association between AL, TT, and HW operates whether or not SS is present. Moreover, the low observed frequency of 4-way co-infections (n = 4) accounts for the low robustness score, further indicating a lack of strong support for this result. Restricting the analysis to males, females, or the most co-infected age group (15–20 year-olds, S2 Appendix) resulted in qualitatively similar outcomes.

### Direct and indirect effects of helminths on cerebral malaria

We tested for both main (direct) and interaction (indirect) effects of helminths on cerebral malaria. We evaluated the main effects of all four helminth species, as well as all 2- and 3-way interactions for AL, TT, and HW. We excluded SS from interaction terms due to its low overall prevalence and low occurrence in helminth co-infections (with just 4 or fewer co-occurrences with any species other than hookworm, S1 Table), leading to insufficient statistical power. This decision was additionally supported by multiple correspondence analysis which indicated that variance in the development of cerebral malaria was associated with less than 8% of the total variance in SS occurrence (all single and co-infections combined) in our study group (S3 Appendix). Our full model testing the effect of helminths on cerebral malaria also included sex and age class as cofactors. Non-significant interaction terms were removed from the full model (AIC = 301) via backwards stepwise selection to achieve the final model shown in Table 3 (AIC = 292). Sex and age class, the direct effects of all four helminth species, and a single two-way interaction between TT and HW were all retained in the final model (Table 3). Both AL and TT were significant and independent predictors of cerebral malaria occurrence,

Table 3. Logistic regression (analysis of deviance and odds ratios) for analysis of helminth co-infection association with the risk of developing cerebral malaria. Statistical significance for each explanatory variable and interaction terms were determined by log-likelihood ratio tests after correcting for all other factors in the model. Interaction terms with *p*-values above 0.1 were removed in reverse step-wise procession.

		Adjusted odds of developing cerebral malaria									
	Estimate	95% CI	Deviance $\chi 2$	<i>p</i> -value							
Intercept	0.324	[0.14 - 0.75]									
Sex (F -> M)	0.921	[0.64 - 1.3]	0.21	0.64							
Age	1.013	[0.99 - 1.04]	0.93	0.33							
A. lumbricoides (AL)	0.211	[0.05 – 0.67]	7.34	0.007	**						
T. trichiura (TT)	0.106	[0.01 - 0.54]	8.63	0.003	**						
Hookworm (HW)	0.632	[0.26 - 1.39]	1.25	0.26							
S. stercoralis (SS)	0.839	[0.28 - 2.20]	0.12	0.73							
TT x HW	12.394	[1.57 – 267.58]	5.91	0.015	*						
AL x TT	-	-	-	-							
AL x HW	-	-	-	-							
AL x TT x HW	-	-	-	-							

\* p<0.05

\*\* p<0.01

https://doi.org/10.1371/journal.pntd.0006483.t003





#### (a) Severe malaria outcomes among helminth-infected patients



**Fig 1.** Malaria severity as (a) the proportion of cerebral outcomes for hyperparasitemic *P. falciparum* malaria cases (incidence) across patients with and without infection by helminths identified to be associated with clinical outcome: *A. lumbricoides* (AL), *T. trichiura* (TT), and hookworm (HW). Helminth infection statuses are mutually exclusive. The number of patients with each (co-)infection status is shown in parentheses above the error bars. Error bars represent 95% confidence interval of the proportion. Pie charts illustrate the relative contribution of each (co-)infection status to (b) the overall number of hyperparasitemic cases and (c) those that developed cerebral malaria.

https://doi.org/10.1371/journal.pntd.0006483.g001

and both were associated with protection from cerebral malaria (Table 3; Fig 1). Infection by AL reduced the odds of developing cerebral malaria by over 80%, while infection by TT was associated with a 90% reduction in cerebral malaria. HW and SS had no significant direct effects on cerebral malaria; however, HW had a significant indirect effect on the development of cerebral malaria via its interaction with TT. Co-infection by HW significantly reduced the protection offered by TT (OR: 12.39 [1.57 – 267.58], p = 0.015; Table 3). Patients infected with neither TT nor HW developed cerebral malaria at a rate of 30.7%, but this fell to 3.2% for patients infected with TT only. However, when patients were infected with both TT and HW the cerebral malaria infection rate rose to 15.8%, which reflects a 46% reduction in the protection offered by TT (Fig 2).

#### Discussion

We found that infection by *Ascaris lumbricoides* (AL) and *Trichuris trichiura* (TT) were both independently associated with significant reductions in cerebral malaria risk, with TT conferring an estimated 90% protection from severe malaria. However, in addition to these direct effects we found that the interaction between two species, TT and hookworm (HW), also had



Fig 2. Significant interaction between *T. trichiura* and hookworm infection on development of cerebral malaria. Error bars represent 95% confidence intervals around the proportions. Cerebral malaria incidence (proportion of hyperparasiteic *P. falciparum* cases that developed cerebral malaria) is reported here without considering other co-factors or co-infections ( $\chi^2 = 5.56$ , p = 0.018).

https://doi.org/10.1371/journal.pntd.0006483.g002

consequences for the development of cerebral malaria. Specifically, the presence of HW nearly halved the protective effect of TT, revealing an intriguing indirect effect of HW on cerebral malaria potentially manifest through its interaction with TT. While the protective effects of single helminth species on cerebral malaria have been previously described [23,24,30,33], the indirect effect we observed in this study provides new evidence that the outcome of pairwise interactions between parasite species can depend on the presence of other parasites in the system. This new insight highlights how complex interactions occurring between parasite species can have non-intuitive consequences for infectious disease outcomes.

Our observation of a direct protective effect of *Ascaris lumbricoides* on cerebral malaria is in agreement with previous reports for this helminth species [29,30]. However, the association between cerebral malaria and *T. trichiura* has not been previously reported. The protection conferred by some helminths on severe malaria disease has been widely attributed to helminth-mediated reductions in immunopathology [30,34,35]. Although, the specific immunoregulatory mechanisms by which species like *A. lumbricoides* and *T. trichiura* confer protection from cerebral malaria are still not fully understood, the effect likely relates to a shift towards regulatory (Treg) and Th2-type lymphocytes which dampen the immune hyper-activation that underlies cerebral malaria [35,36]. For example, experimental infection with the helminth *Schistosoma mansoni* in baboons was shown to protect against severe malaria outcomes following inoculation with *Plasmodium knowlesi*, and immunological assays revealed that helminth-infected individuals showed fewer signs of overactive inflammatory responses [37].

Intriguingly, the protective effect of *Trichuris trichiura* on cerebral malaria was nearly halved in the presence of hookworm. Unlike many other helminth species, the human hookworm, *Necator americanus*, does not stimulate Th2-type immunity, rather it down-regulates the Th2 response [38]. Since *N. americanus* was likely the most common hookworm in our dataset, it is possible that concomitant hookworm infection dampened the immune-regulatory response triggered by *T. trichiura* thereby reversing some of the protection conferred by this species. Although the mechanistic details by which *Trichuris* and hookworm interact to affect cerebral malaria are largely speculative at this point, our result indicates that the co-occurrence of multiple helminth species can have cryptic and profound implications for disease severity.

The direct and indirect associations we found between helminth infections and malaria severity may have been influenced by a number of other risk factors that our data could not address. For example, variance among patients in malaria exposure history, socio-economic status, education level, or environmental factors that influence risk for both helminth infection and development of cerebral malaria could all conceivably change the interpretation of our results. Further research will be needed to understand how such factors, particularly immuno-logical heterogeneity due to previous malaria exposure, might affect the patterns we describe here. Additionally, further studies in populations from different geographic regions could help elucidate how generalizable these associations are across diverse human and pathogen populations.

Co-infection with multiple helminth species is the norm rather than the exception in most hosts [39]. In our dataset, we identified a strong 3-way association between *A. lumbricoides*, *T. trichiura*, and hookworm, in which co-infection with all three species occurred more frequently than expected by random chance. This observation is in line with results from many other human populations [40–43]. Such positive associations between co-infecting helminths could reflect facilitative interactions between species, or may simply be a by-product of co-transmission given the similarity in the life cycle and transmission mode of soil-transmitted helminths in particular [44,45]. Irrespective of the mechanism underlying positive associations between helminth species, our work suggests that the co-occurrence of certain helminth species, in this case, *Trichuris trichura* and hookworm, can strongly shape the outcome of malaria infection. Given the commonness of multi-species helminth infections, these types of interactive effects of co-infecting helminths on microparasite infections may be more important than is currently recognized.

Indirect, or "context-dependent", effects of environment, resources, or host traits have been repeatedly shown to impact parasite interactions [46,47], including the outcome of co-infection (e.g., [48-50]). Our analysis showing a cryptic indirect effect of one parasite on the interaction between two others suggests that the parasite community is another key "context" that can influence the outcome of within-host parasite interactions. In an empirical example of this, [51] showed that competitive outcomes among co-colonizing *Escherichia coli* strains in sterile culture were significantly altered when the experiments were conducted in the presence of natural human gut microbes. Taken together, these studies highlight the importance of moving from a pairwise to a multi-species perspective for understanding co-infection outcomes. Despite growing acceptance of the need for this host-as-habitat approach [52], fueled by the recent uptick in collection of high-throughput microbiome datasets in relation to a suite of transmissible and non-transmissible diseases, there is no consensus on tractable methods for doing so. Multivariate analyses (in particular, factor analysis and structural equation modeling) can help describe communities of organisms and their relationship with a focal infection. However, methods for identifying precise associations or testing interactions involving multiple species (e.g., those outlined in both [31] and [53,54]) continue to suffer from logistical, statistical, and computational limitations. Our study illuminates a compelling

example of non-intuitive multi-species associations, adding to the mounting evidence that such methods are in great need of adoption for the study of parasite interactions.

The hypothetical framework used to explain interactions between gastrointestinal helminths and malaria has shifted over time from a nutritional one [55], to one based on immunosuppression/immunomodulation, in which all worms are pooled together for analysis [29], and finally to one where a distinction between immune mechanisms and hematological ones is made and analyses distinguish between worms with strong immunomodulatory effects (e.g. *Ascaris*) and those with strong effects on blood loss (e.g. hookworm) [56]. However, while hookworms have been regularly associated with increased malaria incidence [56], the present analysis identifies an even more complex negative effect of this worm on malaria outcomes. Thus, even though the development of a hookworm-specific vaccine could have benefits in terms of reducing overall malaria incidence, broad non-specific deworming programs that are not 100% efficient could have unexpected negative consequences in terms of malaria severity. Disease control programs must therefore consider the parasite community as a whole when evaluating the relative costs and benefits of specific control strategies, particularly in the tropics where high pathogen diversity and co-infections are common.

#### Data, code and materials

The datasets supporting this article have been uploaded as part of the supplementary material.

### Supporting information

S1 Table. Helminth species infection status and cerebral malaria occurrence counts for all hyperparasitemic P. falciparum cases. (PDF)

**S1** Appendix. Quantifying interspecific associations. (PDF)

**S2** Appendix. Associations among helminth species in young adults. (PDF)

**S3 Appendix. Multiple correspondence analysis.** (PDF)

## **Author Contributions**

**Conceptualization:** Jessica L. Abbate, Vanessa O. Ezenwa, Jean-François Guégan, Marc Choisy, Mathieu Nacher, Benjamin Roche.

Data curation: Mathieu Nacher, Benjamin Roche.

Formal analysis: Jessica L. Abbate, Benjamin Roche.

Funding acquisition: Benjamin Roche.

Investigation: Jessica L. Abbate, Vanessa O. Ezenwa.

Methodology: Jessica L. Abbate.

Supervision: Vanessa O. Ezenwa.

Writing – original draft: Jessica L. Abbate.

Writing – review & editing: Vanessa O. Ezenwa, Jean-François Guégan, Marc Choisy, Mathieu Nacher, Benjamin Roche.

#### References

- Griffiths EC, Pedersen AB, Fenton A, Petchey OL, Sheffield S. 2011 The nature and consequences of coinfection in humans. J. Infect. 63, 200–6. (https://doi.org/10.1016/j.jinf.2011.06.005) PMID: 21704071
- Lloyd-Smith JO, Poss M, Grenfell BT. 2008 HIV-1/parasite co-infection and the emergence of new parasite strains. *Parasitology* 135, 795–806. (https://doi.org/10.1017/S0031182008000292) PMID: 18371236
- Diedrich CR, Flynn JL. 2011 HIV-1/Mycobacterium tuberculosis coinfection immunology: How does HIV-1 exacerbate tuberculosis? *Infect. Immun.* 79, 1407–1417. (https://doi.org/10.1128/IAI.01126-10) PMID: 21245275
- Pederson AB, Fenton A. 2006 Emphasizing the ecology in parasite community ecology. TREE 22, 133–139. https://doi.org/10.1016/j.tree.2006.11.005 PMID: 17137676
- Koelle K, Cobey S, Grenfell B, Pascual M. 2006 Epochal evolution shapes the phylodynamics of interpandemic influenza A (H3N2) in humans. *Science* 314, 1898–903. (https://doi.org/10.1126/science. 1132745) PMID: 17185596
- Maizels RM, Yazdanbakhsh M. 2003 Immune regulation by helminth parasites: cellular and molecular mechanisms. Nat. Rev. Immunol. 3, 733–744. (https://doi.org/10.1038/nri1183) PMID: 12949497
- Alizon S, de Roode JC, Michalakis Y. 2013 Multiple infections and the evolution of virulence. *Ecol. Lett.* 16, 556–567. (https://doi.org/10.1111/ele.12076) PMID: 23347009
- Goenaga S, Kenney JL, Duggal NK, Delorey M, Ebel GD, Zhang B, Levis SC, Enria DA, Brault AC. 2015 Potential for co-infection of a mosquito-specific flavivirus, nhumirim virus, to block west nile virus transmission in mosquitoes. *Viruses* 7, 5801–5812. (https://doi.org/10.3390/v7112911) PMID: 26569286
- Griffiths EC, Fairlie-Clarke K, Metcalf CJE, Graham AL. 2015 Bottom-up regulation of malaria population dynamics in mice co-infected with lung-migratory nematodes. *Ecol. Lett.* 18, 1387–1396. (https:// doi.org/10.1111/ele.12534) PMID: 26477454
- Seabloom EW, Borer ET, Keding AE, Lacroix C, Mitchell CE, Mordecai EA, Power AG. 2015 The community ecology of pathogens: coinfection, coexistence and community composition. *Ecol. Lett.* 18, 401–415. (https://doi.org/10.1111/ele.12418) PMID: 25728488
- Eldholm V, Rieux A, Monteserin J, Lopez JM, Palmero D, Lopez B, Ritacco V, Didelot X, Balloux F. 2016 Impact of HIV co-infection on the evolution and transmission of multidrug- resistant tuberculosis. *Elife* 5, e16644. (https://doi.org/10.7554/eLife.16644) PMID: 27502557
- Ramiro RS, Pollitt LC, Mideo N, Reece SE. 2016 Facilitation through altered resource availability in a mixed- species rodent malaria infection. *Ecol. Lett.* 19, 1041–1050. (https://doi.org/10.1111/ele.12639) PMID: 27364562
- Graham AL. 2008 Ecological rules governing helminth microparasite coinfection. Proc. Natl. Acad. Sci. 105, 566–570. (https://doi.org/10.1073/pnas.0707221105) PMID: 18182496
- Sih A, Englund G, Wooster D. 1998 Emergent impacts of multiple predators on prey. *Trends Ecol. Evol.* 13, 350–355. (https://doi.org/10.1016/S0169-5347(98)01437-2) PMID: 21238339
- Darling ES, Côté IM. 2008 Quantifying the evidence for ecological synergies. *Ecol. Lett.* 11, 1278– 1286. (https://doi.org/10.1111/j.1461-0248.2008.01243.x) PMID: 18785986
- Morin PJ. 1995 Functional redundancy, non-additive interactions, and supply-side dynamics in experimental pond communities. *Ecology* 76, 133–149. (https://doi.org/10.2307/1940637)
- Preston DL, Henderson JS, Johnson PTJ. 2012 Community ecology of invasions: direct and indirect effects of multiple invasive species on aquatic communities R eports R eports. *Ecology* 93, 1254–1261. (https://doi.org/10.2307/23213754) PMID: 22834365
- Biere A, Bennett AE. 2013 Three-way interactions between plants, microbes and insects. *Funct. Ecol.* 27, 567–573. (https://doi.org/10.1111/1365-2435.12100)
- Romansic JM, Johnson PTJ, Searle CL, Johnson JE, Tunstall TS, Han BA, Rohr JR, Blaustein AR. 2011 Individual and combined effects of multiple pathogens on Pacific treefrogs. *Oecologia* 166, 1029– 1041. (https://doi.org/10.1007/s00442-011-1932-1) PMID: 21400194
- Marra R et al. 2006 Study of the three-way interaction between Trichoderma atroviride, plant and fungal pathogens by using a proteomic approach. *Curr. Genet.* 50, 307–321. (https://doi.org/10.1007/s00294-006-0091-0) PMID: 17008992
- Taylor SL, Wesselingh S, Rogers GB. 2016 Host-microbiome interactions in acute and chronic respiratory infections. *Cell. Microbiol.* 18, 652–662. (https://doi.org/10.1111/cmi.12589) PMID: 26972325
- 22. Salgame P, Yap GS, Gause WC. 2013 Effect of helminth-induced immunity on infections with microbial pathogens. *Nat. Immunol.* 14, 1118–1126. (https://doi.org/10.1038/ni.2736) PMID: 24145791

- Knowles SCL. 2011 The effect of helminth co-infection on malaria in mice: A meta-analysis. Int. J. Parasitol. 41, 1041–1051. (https://doi.org/10.1016/j.ijpara.2011.05.009) PMID: 21777589
- Adegnika AA, Kremsner PG. 2012 Epidemiology of malaria and helminth interaction: a review from 2001 to 2011. *Curr. Opin. HIV AIDS* 7, 221–224. (https://doi.org/10.1097/COH.0b013e3283524d90) PMID: 22418449
- Bradley JE, Jackson JA. 2004 Immunity, immunoregulation and the ecology of trichuriasis and ascariasis. *Parasite Immunol.* 26, 429–441. (https://doi.org/10.1111/j.0141-9838.2004.00730.x) PMID: 15771679
- Christensen N, Nansen P, Fagberni BO, Monrad J. 1987 Heterologous antagonistic and synergistic interactions between helminths and between helminths and protozoans in concurrent experimental infection of mammalian hosts. *Parasitol. Res.* 73, 387–410. (https://doi.org/10.1007/BF00538196) PMID: 3309940
- Christensen NO, Knudsen J, Fagbemi B, Nansen P. 1985 Impairment of primary expulsion of Echinostoma revolutum in mice concurrently infected with Schistosoma mansoni. J. Helminthol. 59, 333–335. (https://doi.org/10.1017/S0022149X00025918) PMID: 3937853
- Lello J, Knopp S, Mohammed K a, Khamis IS, Utzinger J, Viney ME. 2013 The relative contribution of co-infection to focal infection risk in children. *Proc. R. Soc. B Biol. Sci.* 280, 20122813. (https://doi.org/ 10.1098/rspb.2012.2813) PMID: 23303547
- Nacher M, Gay F, Singhasivanon P, Krudsood S, Treeprasertsuk S, Mazier D, Vouldoukis I, Looareesuwan S. 2000 Ascaris lumbricoides infection is associated with protection from cerebral malaria. *Parasite Immunol.* 22, 107–113. (https://doi.org/10.1046/j.1365-3024.2000.00284.x) PMID: 10672191
- Nacher M, Singhasivanon P, Traore B, Vannaphan S, Gay F, Chindanond D, Franetich JF, Mazier D, Looareesuwan S. 2002 Helminth infections are associated with protection from cerebral malaria and increased nitrogen derivatives concentrations in Thailand. *Am J Trop Med Hyg* 66, 304–309. PMID: 12139225
- Vaumourin E et al. 2014 To be or not to be associated: power study of four statistical modeling approaches to identify parasite associations in cross-sectional studies. *Front. Cell. Infect. Microbiol.* 4, 1–11. (https://doi.org/10.3389/fcimb.2014.00001)
- 32. R Core Team. 2015 R: A language and environment for statistical computing.
- Nacher M, Singhasivanon P, Yimsamran S, Manibunyong W, Thanyavanich N, Wuthisen R, Looareesuwan S. 2002 Intestinal helminth infections are associated with increased incidence of Plasmodium falciparum malaria in Thailand. *J Parasitol* 88, 55–58. (https://doi.org/10.1645/0022-3395(2002)088 [0055:IHIAAW]2.0.CO;2) PMID: 12053980
- Hunt NH, Golenser J, Chan-Ling T, Parekh S, Rae C, Potter S, Medana IM, Miu J, Ball HJ. 2006 Immunopathogenesis of cerebral malaria. *Int. J. Parasitol.* 36, 569–582. (https://doi.org/10.1016/j.ijpara. 2006.02.016) PMID: 16678181
- Salazar-Castanon VH, Legorreta-Herrera M, Rodriguez-Sosa M. 2014 Helminth Parasites Alter Protection against Plasmodium Infection. *Biomed Res. Int.* 2014, 913696. (<u>https://doi.org/10.1155/2014/</u> 913696) PMID: 25276830
- Hartgers FC, Yazdanbakhsh M. 2006 Co-infection of helminths and malaria: modulation of the immune responses to malaria. *Parasite Immunol.* 28, 497–506. (https://doi.org/10.1111/j.1365-3024.2006. 00901.x) PMID: 16965285
- Nyakundi RK et al. 2016 Protective effect of chronic schistosomiasis in baboons coinfected with Schistosoma mansoni and Plasmodium knowlesi. Infect. Immun. 84, 1320–1330. (https://doi.org/10.1128/ IAI.00490-15) PMID: 26883586
- Hsieh GC-F et al. 2004 A secreted protein from the human hookworm Necator americanus binds selectively to NK cells and induces IFN-gamma production. J. Immunol. 173, 2699–2704. (https://doi.org/10. 4049/jimmunol.173.4.2699) PMID: 15294988
- Petney TN, Andrews RH. 1998 Multiparasite communities in animals and humans: frequency, structure and pathogenic significance. Int. J. Parasitol. 28, 377–393. PMID: 9559357
- Booth M, Bundy DAP, Albonico M, Chwaya HM, Alawi KS, Savioli L. 1998 Associations among multiple geohelminth species infections in schoolchildren from Pemba Island. *Parasitology* 116 (Pt 1, 85–93. (https://doi.org/10.1017/S003118209700190X)
- Needham C, Kim HT, Hoa NV, Cong LD, Michael E, Drake L, Hall A, Bundy DAP. 1998 Epidemiology of soil-transmitted nematode infections in Ha Nam Province, Vietnam. *Trop. Med. Int. Heal.* 3, 904–912. (https://doi.org/10.1046/j.1365-3156.1998.00324.x)
- Brooker S, Miguel E, Moulin S, Luoba A, Bundy D, Kremer M. 2000 Epidemiology of single and multiple species helminth infections among schoolchildren in Busia district, Kenya. *East Afr. Med. J.* 77, 279– 282. (https://doi.org/10.4314/eamj.v77i3.46613)

- Raso G et al. 2004 Multiple parasite infections and their relationship to self-reported morbidity in a community of rural Côte d'Ivoire. Int. J. Epidemiol. 33, 1092–1102. (https://doi.org/10.1093/ije/dyh241) PMID: 15256525
- Forrester JE, Scott ME, Bundy DAP, Golden MHN. 1988 Clustering of Ascaris lumbricoides and Trichuris trichiura infections within households. Trans. R. Soc. Trop. Med. Hyg. 82, 282–288. (https://doi. org/10.1016/0035-9203(88)90448-8) PMID: 3188157
- Bethony J, Brooker S, Albonico M, Geiger SM, Loukas A, Diemert D, Hotez PJ. 2006 Soil-transmitted helminth infections: ascariasis, trichuriasis, and hookworm. *Lancet* 367, 1521–1532. (https://doi.org/ 10.1016/S0140-6736(06)68653-4) PMID: 16679166
- Chamberlain SA, Bronstein JL, Rudgers JA. 2014 How context dependent are species interactions? Ecol. Lett. 17, 881–890. (https://doi.org/10.1111/ele.12279) PMID: 24735225
- Vale PF, Wilson AJ, Best A, Boots M, Little TJ. 2011 Epidemiological, evolutionary, and coevolutionary implications of context-dependent parasitism. *Am. Nat.* 177, 510–21. (<u>https://doi.org/10.1086/659002</u>) PMID: 21460572
- Laine AL. 2011 Context-dependent effects of induced resistance under co-infection in a plant-pathogen interaction. *Evol. Appl.* 4, 696–707. (https://doi.org/10.1111/j.1752-4571.2011.00194.x) PMID: 25568016
- Gorsich EE, Ezenwa VO, Jolles AE. 2014 Nematode-coccidia parasite co-infections in African buffalo: Epidemiology and associations with host condition and pregnancy. *Int. J. Parasitol. Parasites Wildl.* 3, 124–134. (https://doi.org/10.1016/j.ijppaw.2014.05.003) PMID: 25161911
- Lange B, Reuter M, Ebert D, Muylaert K, Decaestecker E. 2014 Diet quality determines interspecific parasite interactions in host populations. *Ecol. Evol.* 4, 3093–3102. (https://doi.org/10.1002/ece3.1167) PMID: 25247066
- de Muinck EJ, Stenseth NC, Sachse D, vander Roost J, Rønningen KS, Rudi K, Trosvik P. 2013 Context-Dependent Competition in a Model Gut Bacterial Community. *PLoS One* 8, 1–16. (https://doi.org/ 10.1371/journal.pone.0067210) PMID: 23922635
- Costello EK, Stagaman K, Dethlefsen L, Bohannan BJM, Relman D a. 2012 The application of ecological theory toward an understanding of the human microbiome. *Science* 336, 1255–62. (https://doi.org/ 10.1126/science.1224203) PMID: 22674335
- Fenton A, Knowles SCL, Petchey OL, Pedersen AB. 2014 The reliability of observational approaches for detecting interspecific parasite interactions: Comparison with experimental results. *Int. J. Parasitol.* 44, 437–445. (https://doi.org/10.1016/j.ijpara.2014.03.001) PMID: 24704058
- Fenton A, Viney ME, Lello J. 2010 Detecting interspecific macroparasite interactions from ecological data: Patterns and process. *Ecol. Lett.* 13, 606–615. (https://doi.org/10.1111/j.1461-0248.2010.01458. x) PMID: 20529102
- Murray J, Murray A, Murray M, Murray C. 1978 The biological suppression of malaria: an ecological and nutritional interrelationship of a host and two parasites. *Am. J. Clin. Nutr.* 31, 1363–1366. <u>https://doi.org/10.1093/ajcn/31.8.1363</u> PMID: 354372
- Nacher M. 2011 Interactions between worms and malaria: good worms or bad worms? *Malar. J.* 10, 259. (https://doi.org/10.1186/1475-2875-10-259) PMID: 21910854